

**Reviewer's code:** 00504545

**SPECIFIC COMMENTS TO AUTHORS**

This review is very interesting and complete. Describe very exhaustively the long list of potential drugs that can induce acute pancreatitis and the possible mechanisms involved in this process. The description is well organized by different categories and the diagnostic criteria well comprehensive. This list of references is wide and complete including a total of 149 that support the great clinical interest for this subject. The Figures and Tables are very informative.

*Authors Response: Thank you for these kind words. We hope the readership enjoys the manuscript and finds it clinically interesting and informative as well.*

**Reviewer's code:** 03720781

**SPECIFIC COMMENTS TO AUTHORS**

This article reviewed the diversity of DIP and the potential mechanisms of pancreatic injury. It is valuable to help the clinicians to raising awareness of the nature and magnitude of DIP. But there is a small slip that authors made that table1 confused with table2.

*Authors Response: Thank you for pointing out this oversight. We have corrected the table order and it is now reflected appropriately in the revised manuscript.*

**Reviewer's code:** 03000523

**SPECIFIC COMMENTS TO AUTHORS**

Although the information in the manuscript are not original and are mostly well known the authors presents an interesting overview of the current knowledge. I find the tables useful for the clinical practice.

Authors Response: Thank you for these kind comments and for graciously reviewing our manuscript.